Author's response to reviews

Title: Depression symptoms and cognitive function among individuals with advanced HIV infection initiating HAART in Uganda.

Authors:

Noeline Nakasujja (drnoeline@yahoo.com)
Richard L Skolasky (rskolasky@jhmi.edu)
Seggane Musisi (segganemusisi@yahoo.ca)
Peter Allebeck (Peter.Allebeck@ki.se)
Kevin Robertson (kevinr@neurology.unc.edu)
Allan Ronald (aronald@ms.umanitoba.ca)
Elly Katabira (katabira@imul.com)
David B Clifford (cliffordd@neuro.wustl.edu)
Ned Sacktor (sacktor@jhmi.edu)

Version: 7 Date: 31 May 2010

Author's response to reviews: see over
30/05/2010

Dear Scientific Editor,

BMC series journals

Re: Response to concerns raised by the BMC editorial Production Team on MS: 1908145036298464- Depression symptoms and cognitive function among individuals with advanced HIV infection initiating HAART in Uganda.

Thank you for forwarding the formatting changes which are required on the above mentioned manuscript. We address each of them one by one below.

Major revisions (we require the author to make these changes)

Abstract structure: the abstract structure must be background, Methods, Results, Conclusion; please remove the 'objective' title.

Response: We have removed the objective sub title.

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].'

Currently, for example, reference 21 is cited before references 19-20, please amend this.

Response: The references have been edited to appear in order after the text. There are no superscripts in the references.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Table 1: please edit this so that vertical lines are not visible.

Response: The vertical lines in the table have been made invisible.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
Response: We have cropped the figure following the instructions on the above web page.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have edited the manuscript for typographical errors.

Sincerely

Noeline Nakasujja
Corresponding author.